ADPS—A Prescreening Tool for Students With Dyslexia in Learning Traditional Chinese

Ka-Yan Fung[®], Kit-Yi Tang, Tze Leung Rick Lui[®], Kuen-Fung Sin, Lik-Hang Lee[®], Huamin Qu[®], *Member, IEEE*, and Shenghui Song[®], *Senior Member, IEEE*

Abstract—Prescreening children for specific learning disabilities, e.g., dyslexia, is essential for effective intervention. With a quick and reliable prescreening result, special education coordinators (SENCOs) can provide students with early intervention and relieve their learning pressure. Unfortunately, due to the limited resources, many students in Hong Kong receive dyslexia assessments beyond the golden period, i.e., under the age of six. To this end, information technology could establish automatic prescreening tools to address this issue. However, dyslexia prescreening for children learning Chinese is challenging due to the lack of soundscript correlation in Chinese. In this article, an automatic dyslexia prescreening system (ADPS) is developed to provide a quick test to identify at-risk children. Through a two-stage approach, we first develop a gamified tool based on linguistic characteristics and then evaluate the result by a comparison study. Results from a pilot test on 30 students with dyslexia and 32 students without dyslexia indicate that the ADPS can effectively distinguish between two groups of students. Furthermore, the interactive design elements can motivate students to conduct the prescreening independently.

Index Terms—Dyslexia, handwriting (HW), phonological awareness (PA), prescreening, Traditional Chinese, vocabulary knowledge (VK), working memory (WM).

I. INTRODUCTION

D YSLEXIA is a prevalent neurodevelopmental disorder, which affects 5–10% of the global population and primarily impacts reading, writing, and spelling [1]. It poses unique challenges for students learning Traditional Chinese due to its logographic writing system without direct sound–script correlation [2]. This lack of phonetic cues creates more difficulties for students with dyslexia to learn Traditional Chinese, resulting in slower reading, decreased motivation, and increased stress,

This work involved human subjects or animals in its research. Approval of all ethical and experimental procedures and protocols was granted by Institutional Review Board under Application No. HREP2022-0276.

Ka-Yan Fung, Huamin Qu, and Shenghui Song are with the Hong Kong University of Science and Technology, Hong Kong (e-mail: kyfungag@ust.hk; huamin@ust.hk; eeshsong@ust.hk).

Kit-Yi Tang, retired, was with the Chinese University of Hong Kong, Hong Kong. She resides in Hong Kong (e-mail: tangkityi6033@gmail.com).

Tze Leung Rick Lui and Kuen-Fung Sin are with the Education University of Hong Kong, Hong Kong (e-mail: rtl-lui@eduhk.hk; kfsin@eduhk.hk).

Lik-Hang Lee is with the Hong Kong Polytechnic University, Hong Kong (e-mail: lik-hang.lee@polyu.edu.hk).

Digital Object Identifier 10.1109/TLT.2024.3384290

which negatively affects their academic performance and social development [3].

Early intervention is crucial for students with dyslexia, as it can mitigate the adverse effects on learning and reduce the risk of long-term academic and socioemotional consequences [4]. Research has demonstrated that interventions provided before age six are most effective [2], as this is a critical period for language development. Early intervention strategies, such as tailored instruction, multisensory teaching approaches, and targeted support, can significantly improve the reading, writing, and comprehension skills of students with dyslexia [5]. Moreover, early identification and intervention can boost self-esteem, enhance motivation, and foster a positive learning environment for the concerned students [6].

Unfortunately, limited resources, lack of awareness, and inadequate educator training [7] hinder timely assessments and interventions for students with dyslexia. In addition, the existing assessment process is often time consuming and costly, leading to delayed diagnosis and support beyond the critical period for early intervention [8]. The scarcity of standardized assessment tools tailored for Traditional Chinese learners worsens the challenge of identifying dyslexia accurately and efficiently [9]. Consequently, many students with dyslexia struggle academically and emotionally without the necessary support to overcome their learning difficulties [10].

Information technology has the potential to revolutionize the prescreening process for dyslexia, making it more accessible, efficient, and accurate. By leveraging advancements in data analytics and human-computer interaction techniques, automatic prescreening tools can quickly and effectively identify students at risk of dyslexia [11]. These tools can be easily integrated into educational settings, enabling timely interventions and support from educators and special education coordinators (SENCOs). Furthermore, embedding the gamified environment into the prescreening tools can enhance student engagement and motivation, promoting self-directed learning and reducing the stigma associated with dyslexia assessment [12]. Finally, integrating information technology in the prescreening process can help narrow the gap between early identification and intervention for students with dyslexia, improving their academic outcomes and overall well-being. However, the existing tools cannot tackle the above issues for students learning Traditional Chinese because the prescreening is either done manually (difficult to scale) or partially language dependent (does not cover the linguistic characteristics important for Chinese) and sometimes even language

Manuscript received 26 June 2023; revised 13 December 2023 and 9 March 2024; accepted 20 March 2024. Date of publication 2 April 2024; date of current version 23 April 2024. This work was supported by the Hong Kong University of Science and Technology Startup Fund under Grant R9249. (*Corresponding author: Ka-Yan Fung.*)

^{1939-1382 © 2024} IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See https://www.ieee.org/publications/rights/index.html for more information.

Authorized licensed use limited to: Hong Kong University of Science and Technology. Downloaded on May 23,2024 at 17:01:28 UTC from IEEE Xplore. Restrictions apply.

independent. Therefore, there is a critical need to develop an automatic prescreening system that can rapidly identify students at risk of dyslexia, enabling timely intervention and support.

In this article, we propose the automatic dyslexia prescreening system (ADPS), a game-based tool designed to identify students at risk of dyslexia quickly and effectively while learning Traditional Chinese. This study aims to address the research question: How effective is the developed gamified tool in prescreening students with dyslexia in Traditional Chinese?

The study contributes to both the theoretical and practical aspects of dyslexia identification and support. The ADPS employs a two-stage approach incorporating linguistic characteristics [i.e., orthographic awareness (OA), morphological awareness (MA), phonological awareness (PA), handwriting (HW), working memory (WM), and vocabulary knowledge (VK)] and interactive design elements, tackling the challenges faced by students with dyslexia in learning Traditional Chinese language acquisition. The first stage focuses on game development based on the design of learning model (DLM) and linguistic characteristics of dyslexia symptoms, and the second stage provides tests to students (a control group and an experimental group) for a more in-depth analysis of their performance. The gamified nature of the ADPS ensures a user-friendly experience, promoting student engagement while reducing the stigma associated with dyslexia assessments. The work evaluates the effectiveness of the ADPS in distinguishing between students with/without dyslexia in the context of learning Traditional Chinese. The results demonstrate the tool's potential as an efficient, accurate, and engaging prescreening method for early dyslexia identification. In addition, our findings contribute to the understanding of the unique population in Hong Kong, who speak Cantonese and write Traditional Chinese, which is distinguishable from the people in Guangzhou (Written: Simplified Chinese and Oral: Cantonese) and Taiwan (Written: Traditional Chinese and Oral: Mandarin). In conclusion, this study offers valuable insights into developing and implementing a game-based prescreening tool for dyslexia in Traditional Chinese. The main contributions of this work are as follows.

- We developed the ADPS, an efficient prescreening tool for identifying students with dyslexia in Traditional Chinese, utilizing six linguistic characteristics.
- With user-friendly design elements, the ADPS can engage students and enable them to complete prescreening independently without technology literacy issues.
- It was shown that transposition may not be an effective indicator for prescreening students with dyslexia, as it is uncommon in Chinese characters.
- 4) The study revealed that Chinese students with dyslexia generally struggle with VK, MA, and OA. Educators can assist these students by focusing on building a solid vocabulary foundation.

The rest of this article is organized as follows. Section II reviews the related works on nonlinguistic-based and linguistic-based prescreening tools. Section III describes the design principles (see Section III-A) and linguistic characteristics (see Section III-B) in the development stage, discusses the gamified environment (see Section III-C), and overviews the user study in the testing stage (see Section III-D). Section IV

analyzes the prescreening result and reports the observations. Section V discusses the prescreening performance, linguistic characteristics, gamified environments and motivation, and limitations. Finally, Section VI concludes this article.

II. LITERATURE REVIEW

A. Dyslexia and Its Characteristics

Dyslexia is a neurodevelopmental disorder affecting an individual's ability to read, write, and spell [13]. Critical characteristics of dyslexia include phonological processing difficulties, which hinder decoding written words and mapping sounds to letters [14]; reading difficulties, resulting in slower and less accurate reading; spelling and writing challenges due to trouble connecting sounds to appropriate letters [15]; and WM and executive function deficits that affect overall academic performance and daily life [1]. Dyslexia manifests differently among individuals and is unrelated to intelligence or motivation [16]. Early identification and intervention can help individuals with dyslexia develop compelling reading and writing strategies, enabling academic and professional success.

B. Challenges of Learning Traditional Chinese by Students With Dyslexia

Learning Traditional Chinese poses unique challenges for students with dyslexia due to its logographic writing system, which lacks direct sound–script correlation [17]. Without the phonetic cues in alphabetic languages, students with dyslexia struggle to recognize and memorize characters, leading to a slower reading pace and limited vocabulary development [18]. These challenges exacerbate the inherent difficulties associated with dyslexia, making it even more crucial for educators to provide interventions.

C. Existing Assessment Methods for Dyslexia

The existing assessment methods for dyslexia primarily focus on identifying the underlying cognitive and linguistic challenges associated with the disorder. Table I shows the existing assessments, which can be categorized based on the concerned language, such as Traditional Chinese, Simplified Chinese, and alphabets, including Spanish, English, and Bahasa Malay.

1) Traditional Chinese Scripts: Existing works on dyslexia prescreening for Traditional Chinese scripts have diversified focuses, such as OA and MA. For example, Cheng-Lai et al. [19] measured the relationships between orthographic knowledge and perceptual-motor skills with Traditional Chinese characters. The results revealed that: 1) HW speed is related to deficits in rapid automatic naming (RAN) and 2) visual-motor integration is reflected by the number of characters exceeding the grid limits and the variability of character size. McBride-Chang et al. [20] measured cognitive skills, including syllable awareness, tone deletion, RAN, visual skills, and MA. It was shown that word recognition is linked to MA, tone deletion, and visual skills. Moreover, Wong et al. [21] discovered that at-risk students with dyslexia are slower in RAN and weaker in MA and Chinese character recognition than students without dyslexia.

TABLE I
EXISTING PRESCREENING TOOLS

Item	Author	Aim	Language	GB	OA	MA	PA	HW	WM	VK
1	Lam et al. [30]	To assess the Chinese hand- writing performance of stud- ents with/without dyslexia	Traditional Chinese	х	х	х	х	V	х	х
2	Mcbride et al. [24]	To investigate copying skills in relation to word reading and writing in Chinese child- ren with/without dyslexia.	Traditional Chinese	х	V	V	х	v	х	х
3	Cheng-Lai et al. [19]	To measure the relationships with orthographic knowledge and perceptual-motor skills with Traditional Chinese characters.	Traditional Chinese	х	v	х	х	V	x	х
4	McBride-Chang [20]	To measure the cognitive skills, including syllable awareness, tone deletion, RAN, visual skill, and morphological awareness.	Traditional Chinese	х	х	V	v	х	v	х
5	Lin et al. [23]	To investigate the prevalence rate of children with dyslexia, and to evaluate their Chinese reading ability.	Simplified Chinese	х	V	v	v	х	х	V
6	Shu et al. [31]	To understand Chinese reading development and impairment with nine cognitive constructs.	Simplified Chinese	х	х	V	v	х	х	V
7	Huang et al. [22]	To develop a Chinese Reading Ability Test for Chinese primary students who speak Mandarin.	Simplified Chinese	х	v	V	v	х	v	x
8	Rello et al. [32]	To design an online gamified test and a predictive machine model for easier dyslexia learn- ing detection.	Spanish	V	v	v	v	x	v	x
9	Rauschenberger et al. [33]	To screen students with dys- lexia using new indicators related to motor skills.	Spanish	V	х	х	х	х	x	x
10	Rello et al. [34]	To observe how people interact with a linguistic computer-based game.	English	v	v	v	v	х	x	x
11	Ekhsan et al. [29]	To develop a computer-based screening tool by implementing multimedia elements that suitable for dyslexic students.	Bahasa Malaysia	v	v	x	v	x	x	x
12	ADPS (Our work)	To provide a quick test to iden- tify at-risk children based on linguistic characteristics with gamified tool based.	Traditional Chinese	v	v	v	v	v	v	v

The acronyms GB, OA, MA, PA, HW, WM, and VK denote gamified-based, orthographic awareness, morphological awareness, phonological awareness, handwriting, working memory, and vocabulary knowledge, respectively.

2) Simplified Chinese Scripts: Most of the existing works on dyslexia prescreening for Simplified Chinese scripts focused on VK and MA. For example, Huang et al. [22] developed a Chinese Reading Ability Test for Mandarin-speaking primary students. The test included character recognition, PA (i.e., onset and rhyme), MA (i.e., five Chinese compounding structures), RAN (i.e., numbers), OA (i.e., noncharacter recognition), and reading ability measures (i.e., word reading fluency). Lin et al. [23] developed a screening test to measure students' performance in vocabulary, Chinese reading ability, and audiovisual integration, highlighting that simplified Chinese scripts are more prevalent than Traditional Chinese scripts due to the complexity of Traditional Chinese characters. Some existing works also considered HW practice, as it may elaborate on orthographic representation [24].

Relationship between two scripts: Up to 60% of frequently used Traditional and Simplified Chinese characters share the same form [25], such as % (lit. big), and both the scripts follow the same writing rules. Therefore, the HW, copying

skills, and space concepts could be adopted by two scripts. However, some Simplified Chinese characters may present challenges in maintaining consistent orthographic representations compared to their traditional counterparts. For example, the Simplified Chinese character for "love" (爱) lacks the "heart" (心) component in the Traditional Chinese character (愛). In addition, multiple Traditional Chinese characters are often Simplified into a single Simplified Chinese character, leading to confusion about the meaning of words [26]. For example, the Simplified Chinese character "发" is the same in the words "头发" (lit. hair) and "发财" (lit. get rich). However, the original forms in Traditional Chinese are different, namely, "頭髮" (lit. hair) and "發財" (lit. get rich). Therefore, when a Simplified character is recovered from its traditional counterpart, it is easier to identify its meaning. However, students with dyslexia may encounter difficulties in language-relevant content, making dyslexia prescreening with language-relevant content more suitable in the context of Traditional Chinese [27].

3) Alphabets: Prominent literature on dyslexia prescreening for alphabet languages often adopts a gamified- based approach and is less dependent on language content. For example, Rauschenberger et al. [28] screened students with dyslexia using a web game and language-independent content and highlighted that nonlanguage content could facilitate early intervention for pre-readers with no language skills. In addition, Ekhsan et al. [29] developed a multimedia application to screen students with dyslexia. They emphasized that multimedia elements are suitable for prescreening students with dyslexia and provide an alternative to manual screening.

In summary, these assessment methods provide valuable insights into how students' strengths and weaknesses in specific areas can be utilized for dyslexia screening, such as HW performance, copying skills, VK, MA, RAN, and vocabulary skills [30], [31], [23]. The multimedia approach could also effectively engage children's concentration while providing an enjoyable and exciting experience.

D. Gamified Environment and Its Benefits

A gamified environment with an interactive and multimedia approach has been discovered to effectively capture children's attention while offering an enjoyable and thrilling experience [35], [36]. The gamified environment is an innovative educational approach incorporating game elements and design principles, such as performance measurement (badges, level, and dashboards), novelty (new game elements), and sensation (visual or sound stimulation) [37], [38], into the learning process, aiming to increase engagement, motivation, and learning outcomes [39], [40]. In addition, a gamified environment can create a sense of enjoyment for students during prescreening, leading to the unconscious acquisition of knowledge and skills [41]. Furthermore, incorporating animated effects and interactive designs can enhance the effectiveness of the prescreening, as they captivate and inspire students [42]. By integrating interactive experiences, the gamified environment could encourage active participation and increase motivation [43], [44]. Also, intrinsic motivation can lead to improved retention and consolidation of knowledge [45].

E. Limitations of the Existing prescreening Tools

The existing prescreening tools for dyslexia aid in early identification of at-risk students, facilitating timely interventions, and raising awareness between educators and parents. These tools often incorporate standardized tests, questionnaires, or computer-based assessments focusing on dyslexia-related skills, such as PA, decoding, and reading fluency. By identifying potential challenges early, educators can implement targeted support strategies to address students' specific needs, improving their academic performance and overall well-being.

However, they fail to capture all the characteristics of dyslexia, including HW, OA, MA, PA, WM, and VK (i.e., six linguistic characteristics). Also, some traditional assessments rely on the expertise of trained professionals and are human resource intensive. Limited access to resources and funding can also hinder the adoption and implementation of prescreening tools. Consequently, it is essential to develop more inclusive, accessible, and accurate prescreening tools to support better the early identification and intervention efforts of students with dyslexia. Furthermore, a gamified environment can enhance engagement and motivation in the prescreening tests.

F. Improvement Upon Existing Methods

This work improves the existing approaches by incorporating six key linguistic skills with gamified environments for identifying dyslexia in the Chinese context. First, the tool measures Chinese OA, focusing on character recognition. Second, it assesses MA in Chinese by manipulating morphemes. Third, we evaluate PA using Cantonese, the phonetic system prevalent in Hong Kong, to cater to local requirements. Fourth, the proposed tool examines Traditional Chinese HW skills, concentrating on stroke formation, spacing, and character writing, which may reveal underlying dyslexic traits. Furthermore, we assess WM within the context of Chinese language processing, considering the unique demands of recognizing similar characters and reading sentences. Finally, the proposed tool evaluates Chinese VK, addressing word recognition and formation challenges that could indicate dyslexic traits.

III. APPLICATION DESIGN

This study has two stages. In the development stage, experts from different fields collaborated to develop the gamified prescreening tool to bridge the gap between engineering and education. The research team comprises university professors specializing in dyslexia, education research, and integrated system design. The objective is to develop a novel gamified tool, the ADPS, to prescreen at-risk students with dyslexia. The design of ADPS was based on an empirical analysis of errors. Specifically, students' Chinese writings were collected, and their mistakes were annotated with general linguistic characteristics (details will be discussed in Section III-B), including OA, MA, PA, HW, WM, and VK. These mistakes reflect specific difficulties associated with dyslexia and are utilized for dyslexia prescreening.

A. Development Stage: Design Principles

Gamification has demonstrated evidence of great potential in educational technology. Gamification is becoming increasingly popular, particularly in creating gamified designs that motivate students to learn [46]. According to Huang and Soman [47], gamified designs directly affect motivation, which is crucial for students' effective use of learning methods [48]. Saputra and Risqi [35] designed a learning model—DLM—to improve students' learning motivation by using gamification elements. The design principles utilized in this study are as follows.

1) Design of Learning Model: The DLM consists of three phases, including Phase 1: motivational affordances (Game elements), Phase 2: psychological outcomes, and Phase 3: behavioral outcomes. Several motivational affordances are represented by game elements, including: 1) story, clear goals, and levels; 2) points, rewards, and feedback; and 3) achievement/badges. According to the DLM, all the game elements need to follow a



Fig. 1. Game (Set 2, Game 5) comprises a story and clear goals. The red boxes indicate how many questions students need to complete and how many questions are left. The animation of "rabbit eats carrots" notifies students that they have finished a task.



Fig. 2. Radar images indicate the individual performance for each linguistic factor. One radar image corresponds to students' performance in one prescreening test.

consistent set of rules, aiming to produce specific psychological effects. The primary objective is to foster engagement, which generates interest in playing the games. Once this goal is accomplished, the following objectives are to promote enjoyment and motivation, encouraging students to participate in the learning process. The details of the three phases and their application in this study are as follows.

Phase 1: motivational affordances form the foundation of students' learning experience. It comprises a story, clear goals, and difficulty levels that help students comprehend the learning objectives and context and choose a suitable difficulty level. This information is intended to capture students' attention and generate their interest in the subject [49]. Each game element is designed to create specific psychological effects. An intriguing and captivating plot can capture students' attention, promoting engagement and enjoyment. Clear goals and an exciting story can motivate students to achieve the learning objectives [50]. Different difficulty levels are designed to build learning momentum, and students experience a sense of accomplishment as they progress through all the available levels. Students receive points, rewards, and feedback in Phase 2: psychological outcomes. The points and rewards aim to provide students with continuous encouragement and motivation throughout the learning process [51]. For each minor accomplishment, students will receive a point or reward. Feedback should always be positive, regardless of whether they achieve a high or low score, to prevent negative experiences and motivate students to continue learning. The badge is granted upon achieving specific goals in *Phase 3*: behavioral outcomes. The badge boosts students' morale and inspires them to strive for further accomplishments [52].

The prescreening tool's user interface (UI) design developed in this study was also based on the *DLM*. In *Phase 1*, each game contained a story and distinct objectives. A sample game is illustrated in Fig. 1, where a rabbit locates the appropriate carrot with the same Chinese character as the one on its body. Once the rabbit located the correct carrot, it could consume it joyfully. The game also displayed students' progress at the bottom of the screen, informing students that they had five more questions to complete at this level. In Phase 2, each set of prescreening games included a point-and-reward system. Fig. 2 illustrates four radar images, which indicate the individual performance for each linguistic factor, where one radar image corresponds to students' performance in one set of prescreening games. In Phase 3, students were awarded a badge upon accomplishing particular objectives, such as finishing prescreening games. Fig. 3 showcases the badges, which ranged from copper medals to stars, symbolizing the completion of Sets 1-4. Upon finishing the fourth set, a large trophy appeared, accompanied by celebratory music and a character with clapping animation, to commend and motivate the students.

B. Development Stage: Linguistic Characteristics

This section annotates the popular mistakes encountered by students with dyslexia in six linguistic characteristics that are associated with dyslexia, including: 1) OA; 2) MA; 3) PA; 4) HW; 5) WM; and 6) VK.

1) Orthographic Awareness: OA in Chinese refers to understanding conventions and rules for Chinese characters [53], which plays an essential role in Chinese reading and is a core deficit of dyslexia [54]. OA in English pertains to recognizing letter patterns, whereas in Chinese, it involves understanding the rules and conventions for Chinese characters. It enables us to recognize familiar letter patterns. For example, an individual with good OA will recognize "down" in "rundown." Chinese orthography has graphical and functional properties, with radicals serving the purpose of conveying pronunciation and meaning. The radicals also convey the pronunciation and



Fig. 3. Badges demonstrate students' progress. After completing a prescreening test, students would receive a badge. After students complete the four tests, they are awarded four badges and a trophy. This is a gamified environment to encourage students' active participation. (a) Bronze Award. (b) Silver Award. (c) Gold Award. (d) Grand Award. (e) A Champion Trophy.

meaning of the characters they are a part of [55]. For example, the compound character 晴 (lit.: cloudless; Cantonese: cing4) has a left-right structure and consists of two components: 日 (lit.: day; Cantonese: jat6) and 青 (lit.: blue; Cantonese: cing1). The left component 日 is a semantic radical that indicates the related meaning of the character, and the component on the right 青 is a phonological radical that cues the sound (i.e., cing1 to cing4). Research shows that awareness strongly correlates with Chinese literacy development, particularly in reading proficiency [56] and comprehension [57]. Therefore, incorporating this awareness in the prescreening tool can be beneficial.

2) Morphological Awareness: MA involves reflecting on and manipulating the smallest meaningful units in language, and it is essential for language development [58]. Elbro and Arnbak [59] found that people with dyslexia can read morphologically simple words but have difficulty reading complex words. In English, MA includes derivational (e.g., connect and connection) and inflectional morphology (e.g., go and going), which enriches vocabulary and develops grammatical accuracy [60]. MA uniquely contributes to character recognition, writing, reading fluency, and comprehension in Chinese. Morphemes in Chinese refer to the building blocks of characters, which develop the relationship between form and meaning [31], such as a basketball (籃球, laam4 kau4) and the color blue (藍色, laam4 sik1). A few studies have also explored the relationship between MA and gamified digital environments and show that games can be designed to test MA [58]. Incorporating MA in a gamified digital environment can help students develop the complex relationship between form and meaning.

3) Phonological Awareness: PA refers to a person's sensitivity to the phonological structure of spoken language [61]. Snowling [62] found that deficits in PA are a major cause of reading failure in dyslexia, and phonological difficulties persist throughout the development from preschool to adulthood. In English, PA is closely related to literacy and includes phonemic awareness, the ability to manipulate individual sounds in words. For instance, someone with good PA can distinguish between the words "bay" and "pay." However, Chinese is a logographeme system, which is different from the Latin languages' alphabet system. PA in Chinese involves recognizing and manipulating sound units with multiple levels of phonetic units, such as syllables and phonemes [63]. The syllable deletion method has been successfully used to assess PA in Chinese, where participants are required to say the word without one syllable [64]. For example, participants were required to say 火車站 (fo2 ce1 zaam6) without 火 (fo2), where they were expected to say 車站 (ce1 zaam6). PA is crucial for learning to read [65], particularly in Chinese, where over 80% of characters are semantic-phonetic compounds [66]. Character's

meaning, sound, and structure are essential for recognition and memorization [67]. Identifying mispronunciations and RAN are essential to acquiring this awareness [68], [69]. While some tests have been created for mispronunciation detection, limited research has been done on form–sound matching, word pronunciation and meaning confusion, and identifying characters with similar sounds but different meanings. This study aims to fill this gap by designing PA-based prescreening in a digital learning environment.

4) Handwriting: HW is one of the core symptoms that significantly affect people with dyslexia [70]. The difficulty with HW can manifest in different forms, such as poorer legibility [71], larger script size [72], excessive spacing [13], inconsistent spatial orientation [73], and longer writing time [74]. In English, students are usually required to write words on horizontal grid lines. Some common writing characteristics of students with dyslexia include many spelling errors, inability to complete the written tasks [75], or failure to complete written tasks within a given time [76]. Chinese characters are written within a square grid, and their visual complexity [77] is high due to many radicals and strokes in their structure. Elaboration of HW through orthographic representations and formation of motor memory is a common strategy for mastering Chinese reading [78]. For example, 木 (wood), 林 (forest), and 否 (apricot) share the same component π (wood), but the character structures are different. HW problems in Chinese are more related to dyslexia than in alphabetic languages, such as English [79], as characters with similar components can have different structures.

5) Working Memory: WM is a memory system with limited capacity that involves temporary storage and processing of information [80]. Impaired WM is a defining symptom of dyslexia and can significantly impact individuals with dyslexia [81]. Reading or sentence span tasks are suitable measures of WM function, requiring speech processing and selection from the central executive [82]. Low WM capacity leads to reading, spelling, and dictation deficits in students with dyslexia [83]. In English, students with dyslexia make more spelling, punctuation, and capitalization errors and omit more words in dictation tasks [84]. Similar WM tests have been conducted in Chinese, but the lack of sound-script correspondences between phonology and orthography may not directly affect students' performance [85]. Unlike the previous paper-based tools, we designed a gamified digital prescreening tool to address this gap in the gamified environment.

6) Vocabulary Knowledge: VK is recognized as a strong predictor of dyslexia symptoms in reading fluency [86], which has two dimensions: breadth (how many words are known) [87] and depth (how well the meanings are known) [88]. For example,



Fig. 4. Linguistic characteristics related design of four prescreening tests. Each set covers different linguistic contents, and the level of difficulty increases incrementally. The descriptions of gamified design are given in Section III-C.

if a student looks at a picture of a cat, the student can name it 貓 (maau1) in Cantonese. The depth of knowledge relates to individual lexical aspects (i.e., a rich network of associations around a word) [88]. For example, when a student looks at 貓 (cat), the student can associate it with 小貓 (kitten), 熊貓 (panda), 貓頭鷹 (owl), and 花面貓 (a man with dirty face). Students with dyslexia have difficulties in Chinese VK, as shown by previous studies [89]. However, using a single measure of vocabulary is not enough, as some students may recognize words in a picture format but not in a written form. This study addresses this gap by adopting both the breadth and depth dimensions for prescreening students with dyslexia in a gamified digital environment.

C. Gamified Environments

Fig. 4 shows the practical implementation of the gamified environment. The content design is based on linguistic characteristics, and the UI design is based on the *DLM* (see Section III-A). The prescreening system included four sets of tests, each with ten games. A detailed description of the four sets of games and the corresponding linguistic characteristics to be tested is shown as follows.

Set 1:

- 1) *Game 1* applied the drag-and-drop approach and examined if students could differentiate similar Chinese characters with the same sound (characteristics tested: MA).
- 2) *Game 2* was a matching game and tested students' memory of the same Chinese character (characteristics tested: WM).
- Game 3 utilized the drag-and-drop approach and evaluated students' ability to recognize wrong components (characteristics tested: OA).
- 4) *Game 4* analyzed whether students could identify the Chinese characters by stroke addition (characteristics tested: OA).

- 5) *Game 5* scrutinized if students could distinguish the Chinese characters by stroke deletion (characteristics tested: OA).
- 6) *Game 6* was a reading task to check students' reading fluency (characteristics tested: WM).
- 7) *Game* 7 was a writing test to understand students' stroke concepts by writing one component of a word (characteristics tested: WM).
- 8) *Game* 8 was a pre-dictation. Students were requested to dictate the first word of a vocabulary (characteristics tested: WM).
- Game 9 was a one-word copying test with single (e.g., 木) and left-right (e.g., 好) structures to assess students' space concept (characteristics tested: HW).
- 10) *Game 10* was a three-word copying task with a single and left–right structure to evaluate students' character construction awareness (characteristics tested: HW).

Set 2:

- 1) *Game 1* applied the drag-and-drop approach and examined if students could differentiate similar Chinese characters (characteristics tested: OA).
- 2) *Game 2* was a matching game that gave students visual cues to find the right word from similar Chinese characters (characteristics tested: OA).
- 3) *Game 3* was a matching game and requested students to form a correct vocabulary from similar Chinese characters (characteristics tested: OA).
- 4) *Game 4* tested if students could distinguish the correct characters from the reversal of Chinese characters (characteristics tested: OA).
- 5) *Game 5* scrutinized if students could distinguish the correct characters from their rotation (characteristics tested: OA).
- 6) *Game 6* was a reading task to check students' reading fluency (characteristics tested: OA).

- Game 7 checked if students could identify Chinese characters with similar sounds but different meanings (characteristics tested: PA).
- 8) *Game* 8 was a post-dictation. Students were requested to dictate the second word of a vocabulary (characteristics tested: OA).
- 9) *Game 9* was a one-word copying test with left-right and up-down (e.g., 吉) structure to assess students' space concept (characteristics tested: HW).
- 10) *Game 10* was a three-word copying task with left–right and up–down structure to evaluate students' word formation awareness (characteristics tested: HW).

Set 3:

- 1) *Game 1* tested students' MA for words with the same sound (characteristics tested: VK).
- Game 2 examined students' ability to identify the transposition of Chinese characters (characteristics tested: OA).
- 3) *Game 3* reviewed students' ability to identify samenature vocabularies (characteristics tested: VK).
- 4) *Game 4* was a matching task to understand students' MA (characteristics tested: MA).
- 5) *Game 5* was a matching task to assess students' vocabulary identification ability (characteristics tested: VK).
- 6) *Game 6* was a task of vocabulary recognition (characteristics tested: PA).
- 7) *Game 7* evaluated students' ability of RAN (characteristics tested: VK).
- 8) *Game 8* was a reading task to check students' reading fluency (characteristics tested: OA).
- Game 9 was a one-word copying test with left-up-down (e.g., 陪) and left-middle-right (e.g., 樹) structure to assess students' space concept (characteristics tested: HW).
- 10) *Game 10* was a three-word copying task with left-updown and left-middle-right structure to evaluate students' word formation awareness (characteristics tested: HW).

Set 4:

- 1) *Game 1* evaluated students' ability to distinguish word meaning (characteristics tested: MA).
- 2) *Game 2* was a task to assess students' vocabulary size (characteristics tested: VK).
- 3) *Game 3* was a task to examine students' ability of vocabulary expansion (characteristics tested: VK).
- 4) *Game 4* reviewed students' ability to do odd-man-out tasks (characteristics tested: VK).
- 5) *Game 5* tested students' ability in form–sound matching (characteristics tested: PA).
- 6) *Game 6* evaluated if students were affected by the pronunciation of words and confused by word meaning and form (characteristics tested: MA).
- 7) *Game* 7 checked students' reading fluency (characteristics tested: OA).
- 8) *Game* 8 was a one-word copying test with semi- (e.g., 包) and full-enclosed (e.g., 固) structure to assess students' space concept (characteristics tested: HW).

- Game 9 was a three-word copying task with a single and left-right structure to evaluate students' word formation awareness (characteristics tested: HW).
- 10) *Game 10* was a task of "Look-say-cover-write-check" to test students' memory (characteristics tested: OA).

Each game consisted of five mini-games and one validation game. The validation game was randomly selected from the five mini-games. All the game contents were in level one, which was based on the study of the Chinese Characters recommended for the subject of Chinese Language in Primary Schools,¹ and designed by a local research team [72]. The difficulty of the four sets of games increases substantially. In addition, the gamified environment employed large icons with all the instructions supporting the text-to-speech (TTS) function, as shown in Fig. 5. The app was developed using Unity and could run on tablets, such as iPad Air.

D. Testing Stage: User Study

A user study was conducted from February 2022 to December 2022. The participants were recruited based on purposeful sampling. This sampling technique is widely used in qualitative research to identify and select informative cases [90]. In this study, students in the control group were without any symptoms of dyslexia, and students in the experimental group were with dyslexia. The criteria for recruiting participants are: 1) studying in grades 1–3; 2) being able to read and write Traditional Chinese characters and speak Cantonese; and 3) having no other medical or physical disabilities that might interfere with their HW ability and reading aloud ability. Also, all the students had experience using tablets.

1) Regulations: This study followed the following regulations: 1) informed consent was obtained from the student's parents before running the experiment; 2) participation was entirely voluntary and based on consent; and 3) the University's Institutional Review Board approved the experimental protocol (Number: HREP2022-0276).

2) *Participants:* The participants were 62 students (27 females and 35 males) aged 7–9 (average age is 8.340 with a 0.655 standard deviation) from a local primary school in Hong Kong. 30 of them have dyslexia, and 32 have no dyslexia identified.

3) Study Method: Each participant was enrolled in the ADPS test for four days and was required to play one set of tests daily. Each test took 30 min, with the difficulty level increasing incrementally.

4) Procedures: Before the test, students had a 5-min faceto-face (F2F) training, instructing them on the function of each button. One instructor provided technical support to two students (as shown in Fig. 6). Then, the instructors assisted students to log in to the app. As shown in Fig. 7, students conducted the prescreening. Prior to the commencement of each game, the app provided a game demo, ensuring that every student was acquainted with the gameplay mechanics. After that, students conducted the prescreening games, and the app uploaded the results and interaction data, such as the start/end times and answer

¹[Online]. Available: https://ephpth.ephhk.com/resource/tools/lcprichi/



Fig. 5. Gamified environment employs large icons with instructions that support the TTS function. This game utilizes the animation of "finding alien" to attract students' attention (Set 2, Game 7).



Fig. 6. Pilot test was conducted in a classroom setting. Participants were working on the prescreening game.



Fig. 7. Overview of the study procedure. The yellow bow indicates the preparation task for participants, while the orange boxes denote device-based user activities primarily in the preparation stage. The core prescreening materials are enclosed within the purple-colored frame, and the rightmost box in blue depicts the meta activity.

choices, to the cloud server. Finally, students could review their performance. After each test, the instructor interviewed students by asking some open-ended questions. The interview typically lasted 5 min and was recorded manually.

IV. DATA ANALYSIS

We constructed a database to gather user data from the prescreening tool for data analysis. The data encompass the correctness rate, answer time, and HW images. Regarding the time spent, the start time is logged when participants initiate the game by clicking the game icon, and the end time is recorded when participants press the completion icon to conclude the game.

A. Evaluation Metrics

Each set consisted of ten games. Each game included five mini-games. Except for HW, all other games had the same weight, i.e., one point per mini-game. The evaluation metrics of HW were shown as follows:

Handwriting: The grading criteria were adapted from [52] and [91], which were used in the same district (see Fig. 8).

- 1) *Label 1 (Noncharacters):* A significant part of compound characters consists of graffiti or only the radicals.
- 2) *Label 2 (Directionality):* A portion of the characters or the entire character is a mirror character.

Dependent Variables	Correct	Incorrect
Label 1: Non-characters. A significant part of compound characters consists of graffiti or only the radicals.	品	
Label 2: Directionality. A portion of the characters or the entire character is a mirror character.	品	
Label 3: Transposition. Characters contain transposed strokes or components.	能	
Label 4: Stroke Insertion/Deletion. Characters have one or more added/deleted strokes.	能	能
Label 5: Broken Stroke Errors. A component or single-component character is divided into multiple parts.	品比月比	
Label 6: Incorrect Stroke Type. An incorrect stroke type is used.	品	
Label 7: Omission of Radicals/Addition of Components. The radical of a character is missing, or an additional component is inserted into the character.	日上	
Label 8: Component Replacement. Components are misplaced.	能	

Fig. 8. HW grading criteria consist of eight common mistakes.

- 3) *Label 3 (Transposition):* Characters contain transposed strokes or components.
- 4) *Label 4 (Stroke insertion/deletion):* Characters have one or more added/deleted strokes.
- 5) *Label 5 (Broken stroke errors):* A component or singlecomponent character is divided into multiple parts.
- 6) *Label 6 (Incorrect stroke type):* An incorrect stroke type is used.
- 7) *Label 7 (Omission of radicals/addition of components):* The radical of a character is missing, or an additional component is inserted into the character.
- 8) Label 8 (Component replacement): Components are misplaced.

B. Results of Four Prescreening Sets

In this section, we discuss the prescreening results. SPSS 29.0.1.0 was utilized to process the data. The descriptive and correlation study for four prescreening sets and the associated descriptive and correlation study variables in linguistic characteristics are provided in Tables II and III, respectively.

To answer the research question, a repeated measures analysis of variance (ANOVA) was conducted to assess the differences between groups, as shown in Table II. Students without dyslexia performed significantly better than those with dyslexia. Both average score and time differences in four sets between the two groups were statistically significant. As the difficulty level of the four prescreening sets increases incrementally, the performance of the two groups over the four sets decreases, as shown in



Fig. 9. Performance comparison between two groups of students. (a) Score overview of four sets of prescreening. Performance of both the groups declined as expected as the difficulty level increased incrementally. (b) Time overview of four sets of prescreening. Both the groups exhibited longer completion time for Sets 3 and 4 compared to Sets 1 and 2. (c) Linguistic characteristic overview of two groups of students. Both the groups faced challenges in handwriting and vocabulary due to script complexity and students' vocabulary size.

 TABLE II

 Descriptive and Correlation Study in Four Sets

Score							eta	squared		Time				eta	squared	
Set	Group	Ν	Μ	SD	F-val	Sig.	η^2	F-val	Sig.	Μ	SD	F-val	Sig.	η^2	F-val	Sig.
1	ND	32	78.637	4.71	43.335	**	0.419	43.337	**	439.59	147.893	10.429	**	0.148	10.429	**
	Dys	30	69.95	5.663						551.47	122.744					
2	ND	32	76.538	11.355	7.482	***	0.111	7.483	**	462.47	124.899	8.032	**	0.118	8.032	**
	Dys	30	69.303	9.289						566.6	163.01					
3	ND	32	74.02	17.761	8.487	**	0.124	8.487	**	678.75	178.768	8.904	**	0.129	8.904	**
	Dys	30	61.443	16.123						830.27	220.075					
4	ND	32	73.203	11.617	15.377	***	0.204	15.378	**	693	189.591	4.86	*	0.075	4.86	*
	Dys	30	60.881	13.116						819.5	258.955					

Note: *** = p < 0.001, ** = p < 0.01, and * = p < 0.05. The acronyms ND, Dys, and Val are non-dyslexia, dyslexia, and value, respectively.

Score								eta	squared	
	Group	Ν	М	Diff.	SD	F-value	Sig.	η^2	F-value	Sig.
OA	ND	32	90.63	29.47%	16.837	17.812	***	0.229	17.812	**
	Dys	30	70		21.496					
MA	ND	32	63.88	41.33%	15.591	25.528	***	0.298	25.528	**
	\mathbf{Dys}	30	45.2		13.335					
PA	ND	32	68.44	23.69%	14.449	12.445	***	0.172	12.445	**
	\mathbf{Dys}	30	55.33		14.794					
HW	ND	32	48.22	6.38%	8.556	1.741	0.192	0.025	1.537	0.22
	Dys	30	45.33		8.656					
WM	ND	32	71.34	13.60%	7.512	19.543	***	0.245	19.457	***
	\mathbf{Dys}	30	62.8		7.703					
VK	ND	32	40.53	38.47%	12.295	15.204	***	0.198	14.831	***
	$_{\rm Dys}$	30	29.27		10.285					
Time								eta	squared	
Time	Group	N	М	Diff.	SD	F-value	Sig.	$\frac{\text{eta}}{\eta^2}$	squared F-value	Sig.
Time OA	Group ND	N 32	M 36.84	Diff. 0.79%	SD 19.286	F-value 0.004	Sig. 0.951	$rac{\text{eta}}{\eta^2}$	squared F-value 0.004	Sig. 0.951
Time OA	Group ND Dys	N 32 30	M 36.84 37.13	Diff. 0.79%	SD 19.286 17.234	F-value 0.004	Sig. 0.951	$\frac{\text{eta}}{\eta^2}$	squared F-value 0.004	Sig. 0.951
OA MA	Group ND Dys ND	N 32 30 32	M 36.84 37.13 232.84	Diff. 0.79% 14.37%	SD 19.286 17.234 74.798	F-value 0.004 2.209	Sig. 0.951 0.142	$\begin{array}{c} \text{eta} \\ \hline \eta^2 \\ 0 \\ 0.036 \end{array}$	squared F-value 0.004 2.209	Sig. 0.951 0.142
Time OA MA	Group ND Dys ND Dys	N 32 30 32 30	M 36.84 37.13 232.84 266.3	Diff. 0.79% 14.37%	SD 19.286 17.234 74.798 101.247	F-value 0.004 2.209	Sig. 0.951 0.142	$ \begin{array}{c} \text{eta} \\ \eta^2 \\ 0 \\ 0.036 \end{array} $	squared F-value 0.004 2.209	Sig. 0.951 0.142
Time OA MA PA	Group ND Dys ND Dys ND	N 32 30 32 30 32 32	M 36.84 37.13 232.84 266.3 117.66	Diff. 0.79% 14.37% 7.07%	SD 19.286 17.234 74.798 101.247 34.339	F-value 0.004 2.209 0.846	Sig. 0.951 0.142 0.361	eta η^2 0 0.036 0.014	squared F-value 0.004 2.209 0.846	Sig. 0.951 0.142 0.361
Time OA MA PA	Group ND Dys ND Dys ND Dys	N 32 30 32 30 32 30 32 30	M 36.84 37.13 232.84 266.3 117.66 125.97	Diff. 0.79% 14.37% 7.07%	SD 19.286 17.234 74.798 101.247 34.339 36.821	F-value 0.004 2.209 0.846	Sig. 0.951 0.142 0.361	eta η^2 0 0.036 0.014	squared F-value 0.004 2.209 0.846	Sig. 0.951 0.142 0.361
Time OA MA PA HW	Group ND Dys ND Dys ND Dys ND	N 32 30 32 30 32 30 32 30 32	M 36.84 37.13 232.84 266.3 117.66 125.97 579.06	Diff. 0.79% 14.37% 7.07% 40.98%	SD 19.286 17.234 74.798 101.247 34.339 36.821 232.285	F-value 0.004 2.209 0.846 8.331	Sig. 0.951 0.142 0.361 *	$ \begin{array}{c} \text{eta} \\ \eta^2 \\ 0 \\ 0.036 \\ 0.014 \\ 0.122 \\ \end{array} $	squared F-value 0.004 2.209 0.846 8.331	Sig. 0.951 0.142 0.361 **
Time OA MA PA HW	Group ND Dys ND Dys ND Dys ND Dys	N 32 30 32 30 32 30 32 30 32 30	M 36.84 37.13 232.84 266.3 117.66 125.97 579.06 816.33	Diff. 0.79% 14.37% 7.07% 40.98%	SD 19.286 17.234 74.798 101.247 34.339 36.821 232.285 398.516	F-value 0.004 2.209 0.846 8.331	Sig. 0.951 0.142 0.361 *	$ \begin{array}{c} \text{eta} \\ \eta^2 \\ 0 \\ 0.036 \\ 0.014 \\ 0.122 \\ \end{array} $	squared F-value 0.004 2.209 0.846 8.331	Sig. 0.951 0.142 0.361 **
Time OA MA PA HW WM	Group ND Dys ND Dys ND Dys ND Dys ND	N 32 30 32 30 32 30 32 30 32 30 32	M 36.84 37.13 232.84 266.3 117.66 125.97 579.06 816.33 822.34	Diff. 0.79% 14.37% 7.07% 40.98% 24.18%	SD 19.286 17.234 74.798 101.247 34.339 36.821 232.285 398.516 217.061	F-value 0.004 2.209 0.846 8.331 14.606	Sig. 0.951 0.142 0.361 * ***	$\begin{array}{c} \text{eta} \\ \eta^2 \\ 0 \\ 0.036 \\ 0.014 \\ 0.122 \\ 0.196 \end{array}$	squared F-value 0.004 2.209 0.846 8.331 14.606	Sig. 0.951 0.142 0.361 ** ***
Time OA MA PA HW WM	Group ND Dys ND Dys ND Dys ND Dys ND Dys ND	N 32 30 32 30 32 30 32 30 32 30 32 30	M 36.84 37.13 232.84 266.3 117.66 125.97 579.06 816.33 822.34 1021.17	Diff. 0.79% 14.37% 7.07% 40.98% 24.18%	SD 19.286 17.234 74.798 101.247 34.339 36.821 232.285 398.516 217.061 190.627	F-value 0.004 2.209 0.846 8.331 14.606	Sig. 0.951 0.142 0.361 * ***	$ \begin{array}{c} \text{eta} \\ \eta^2 \\ 0 \\ 0.036 \\ 0.014 \\ 0.122 \\ 0.196 \\ \end{array} $	squared F-value 0.004 2.209 0.846 8.331 14.606	Sig. 0.951 0.142 0.361 ** ***
Time OA MA PA HW WM VK	Group ND Dys ND Dys ND Dys ND Dys ND Dys ND	N 32 30 32 30 32 30 32 30 32 30 32 30 32	M 36.84 37.13 232.84 266.3 117.66 125.97 579.06 816.33 822.34 1021.17 485.06	Diff. 0.79% 14.37% 7.07% 40.98% 24.18% 3.27%	SD 19.286 17.234 74.798 101.247 34.339 36.821 232.285 398.516 217.061 190.627 133.175	F-value 0.004 2.209 0.846 8.331 14.606 0.202	Sig. 0.951 0.142 0.361 * *** 0.655	$\begin{array}{c} \text{eta} \\ \eta^2 \\ 0 \\ 0.036 \\ 0.014 \\ 0.122 \\ 0.196 \\ 0.003 \end{array}$	squared F-value 0.004 2.209 0.846 8.331 14.606 0.202	Sig. 0.951 0.142 0.361 ** *** 0.655

TABLE III DESCRIPTIVE AND CORRELATION STUDY VARIABLES IN LINGUISTIC CHARACTERISTICS

Note: *** = p < 0.001, ** = p < 0.01, and * = p < 0.05. The acronyms ND, Dys, and Diff. are non-dyslexia, dyslexia, and differences, respectively.



Fig. 10. Score performance of Set 1. Diff. denotes the percentage difference between two groups.



Fig. 11. Score performance of Set 2. Diff. denotes the percentage difference between two groups.

Fig. 9(a). However, the performance of students with dyslexia dropped more significantly than that of those without dyslexia, especially from Sets 2 to 3. Students with dyslexia did poorly in Set 3, because they are weak in VK, and 40% of the questions in Set 3 test VK.

Furthermore, it took longer for students with dyslexia to complete the game. As shown in Fig. 9(b), both the groups spent more time completing Sets 3 and 4 than Sets 1 and 2. The reason is that Sets 3 and 4 involved more diverse questions and focused more on students' VK, which was challenging for both the groups. However, students with dyslexia had a more significant drop in Sets 3 and 4 compared with those without dyslexia.

Surprisingly, the increase in time consumption for both the groups was very close. In addition, we conducted ANOVAs to examine the statistically significant difference between the performance of the two groups. The analysis revealed that students with dyslexia spent much more time finishing the four sets of tests than those without dyslexia, and the result was statistically significant. Therefore, the score and time consumption results indicated that the four sets of prescreening tests could significantly differentiate two groups of students, supporting the research question.

A repeated measures ANOVA was performed to understand how each linguistic characteristic contributes to the differentiation of two groups (refer to Section III-B). The detailed analysis of the performance of the prescreening tests was shown as follows: Set 1 (see Fig. 10), Set 2 (see Fig. 11), Set 3 (see Fig. 12), and Set 4 (see Fig. 13). The analysis is a single-factor ANOVA for the score performance. The *Y*-axis represents the score received, and the *X*-axis represents each game in the tests.

In Set 1 (see Fig. 10), students with dyslexia exhibited a significant weakness in reading fluency (Game 6) and stroke concept (Game 7), which were 29.03% and 56.45% lower than those without dyslexia, respectively. These two games are related to WM. However, the differences were mild in Games 4 and 5 (OA) and Game 10 (HW).

In Set 2 (see Fig. 11), students with dyslexia were significantly weaker in reading fluency (Game 6) and dictation (Game 8), which were 38.78% and 80.10% lower than those without dyslexia, respectively. These two games also belong to WM. However, the differences were mild in Game 2, Game 7, and Game 10, which test OA, PA, and HW, respectively.

In Set 3 (see Fig. 12), students with dyslexia had significant weakness in the same sound word (Game 1), same nature identification (Game 2), MA (Game 4), and vocabulary recognition (Game 6), which were 36.22%, 32.12%, 30.21%, and 37.14% lower than those of without dyslexia, respectively. These four games are associated with VK, OA, MA, and PA, respectively. However, the differences were mild in Game 2 and Game 9, which are about OA and HW.



Fig. 12. Score performance of Set 3. Diff. denotes the percentage difference between two groups.



Fig. 13. Score performance of Set 4. Diff. denotes the percentage difference between two groups.

In Set 4 (see Fig. 13), students with dyslexia showed significant weakness in the confusion of word meaning (Game 1), vocabulary size (Game 2), vocabulary expansion (Game 3), oddman-out (Game 4), and form–sound matching (Game 5), which were 39.55%, 37.71%, 45.70%, 48.45%, and 39.87% lower than those without dyslexia, respectively. These five games are related to MA, VK, and PA. However, the differences were mild in Game 8, corresponding to HW. Overall, the weakness of students with dyslexia covers different linguistic characteristics, as shown in Fig. 9(c).

Overall, the weakness of students with dyslexia covers different linguistic characteristics. As shown in Fig. 9(c), both the groups struggled with HW and VK, which could be attributed to the complexity of the scripts and the students' vocabulary size. However, the disparities between the two groups were more obvious in OA, PA, MA, and VK.

C. Results Regarding Linguistic Characteristics

Table III shows the result regarding linguistic characteristics. The performance of students with dyslexia was far behind those without dyslexia in all the six linguistic characteristics. Except for HW, the differences in other characteristics were statistically significant. Specifically, the differences in MA and VK were the most obvious, and the average scores of students without dyslexia were 41.33% and 38.47% higher than those with dyslexia, respectively. Regarding time consumption, students with dyslexia spent more time on HW and WM than those without dyslexia. The differences were statistically significant. The time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and WM of students with the time consumption differences in HW and the time consumption differences in HW and the time consumption differences in HW and WM of students with the time consumption differences in HW and the time consumption differences in

dyslexia were 40.98% and 24.18% higher than those without dyslexia, respectively.

V. DISCUSSION

The upcoming sections cover several topics, including students' performance in the prescreening tests, their performance in each linguistic characteristic, the potential of gamified environments to motivate students, the significance of the audio recording function, and the challenges encountered while conducting this prescreening test.

A. Performance

In this section, we would like to discuss the linguistic characteristics at which students with dyslexia are weak. Sets 1 and 2 mainly focus on WM and OA. The performance gap between the two groups was relatively small. Sets 3 and 4 are more diverse and cover all the linguistic characteristics listed in Section III-B, and both groups' performance dropped. The performance gap between the two groups was significant, which revealed that students with dyslexia could not master a higher level of Chinese. As shown in Section IV, students with dyslexia could master the fundamental level of Chinese. However, the performance of students with dyslexia declined faster than students without, especially in Sets 3 and 4. Surprisingly, Chinese HW was not very difficult for students with dyslexia, which is different from the observations of previous research [92].

Authorized licensed use limited to: Hong Kong University of Science and Technology. Downloaded on May 23,2024 at 17:01:28 UTC from IEEE Xplore. Restrictions apply.

Although many studies found significant writing difficulties among students with dyslexia [93], [94], our test results revealed that the writing ability gap between the two groups was relatively small. Instead, students with dyslexia showed a larger discrepancy from their peers in other linguistic characteristics, such as WM, OA, PA, and MA. The following section discusses students' performance in each linguistic characteristic.

First, the scores of the linguistic characteristics are quite different between the two groups, except for HW, which is different from prior research [95]. The result revealed that some HW characteristics may exist in English but not Chinese. For example, transposition is a common characteristic in English [96], and students with dyslexia may spell "was" as "saw." However, only rare cases of Chinese characters have transposed forms (e.g., \mathcal{P}) and \mathcal{P}). Therefore, transposition is not a good matrix for prescreening students with dyslexia in Chinese.

Second, the motor skills of young children may not be well developed, and they may not be able to write words neatly. Most young children may face the same HW problem [97], and HW disability may not be the most significant factor in prescreening students with dyslexia. Also, students with dyslexia constitute a diverse group with varying profiles of strengths and weaknesses. While writing difficulties are commonly associated with dyslexia, the severity and manifestation of these difficulties vary among individuals. Some students with dyslexia may exhibit relatively poor WM abilities compared to their writing skills, leading to a smaller writing ability gap but a larger disparity in WM performance compared to their peers.

Third, the results from both Sets 1 and 2 implied that students with dyslexia performed poorly in WM. WM deficit for students with dyslexia is not limited to phonological component dysfunction but also encompasses visual-object and visual-spatial information [98]. However, students with dyslexia did not exhibit severe difficulties in OA in this study, which differs from previous observations [99]. With sufficient visual hints, students with dyslexia could retrieve their memory.

Fourth, our results revealed that students with dyslexia encountered difficulties with higher levels of tests in Sets 3 and 4 in all the linguistic characteristics, especially VK, MA, and PA. As mentioned above, MA develops the complex relationship between form and meaning [100]. At the same time, VK is measured by how many vocabularies are known and how well the meanings are known. Therefore, the performance of MA can predict the performance of VK.

Finally, after three years of the COVID-19 pandemic, the abilities of both the groups of students may have changed. Before the pandemic, students primarily engaged in F2F instruction, which provided ample time for writing and vocabulary acquisition. However, during the pandemic, students shifted to online learning, reducing writing practice for both the groups of students. As a result, both the groups experienced a decline in their writing abilities. On the other hand, they also had more time to listen, speak, and recognize words during the pandemic. The characteristics of dyslexia may be more evident in areas, such as PA, OA, and MA. Therefore, at-risk students with dyslexia may be identified with different indicators.

C. Time Factor

From the data analysis, the increase in time consumption for both the groups was very similar. Previous research has suggested that students with dyslexia spend more time completing tasks [91], [101]. While our findings are consistent with this, we also found that both the groups, regardless of dyslexia status, required more time to complete difficult tasks, and the percentage of additional time needed was comparable.

The insight gained from this finding is that the task's difficulty level, rather than the presence of dyslexia, is the primary factor in determining the amount of time needed to complete it. This challenges the common assumption that students with dyslexia always require more time to complete tasks and highlights the importance of considering task difficulty when assessing students' performance. In addition, the fact that both the groups required a similar percentage of additional time suggests that accommodations for students with dyslexia such as extra time, may not necessarily put them in an unfair situation compared to their peers.

Furthermore, while some students required more time to complete tasks, they were still able to do so. In such cases, it would be premature to conclude that these students have dyslexia. Instead, they may simply require more time to retrieve knowledge from memory. The longer time required for memory retrieval could be due to various factors such as unfamiliarity with the task type or certain vocabulary [102]. With additional training, students can improve their memory retrieval ability and reduce anxiety [103]. Therefore, we cannot rely solely on completion time to prescreen dyslexia; time can be one of several factors to consider.

D. Gamified Environments and Motivation

The results of this study suggest that most students responded positively to the gamified prescreening tool. In particular, students reported enjoying the games and animations used in the tool. This finding is consistent with previous research by Annetta et al. [104], which found that students often perceive gamified activities as "playing games" rather than "working on a test." The positive response to the animations used in the tool suggests that visual and auditory elements can enhance student engagement in learning activities, which is an important consideration for designers of gamified learning platforms. Interestingly, students were able to recall which game they played, indicating that they could differentiate between the different games presented in the tool. This finding suggests that the gamified prescreening tool effectively engages students and promotes their interest in the learning activities. As noted by Garris et al. [105], the challenge of gamification lies in modifying game features to facilitate instruction by engaging the game cycle that fosters self-directed interest without compromising the enjoyable aspects of games.

E. Strengths of ADPS

The ADPS is a tailored prescreening tool to enhance early identification and intervention. The strengths are threefold. First, the ADPS addresses the unique challenges faced by students with dyslexia in learning Traditional Chinese, filling a gap in the assessment tools. Second, by providing a quick and effective prescreening method, the ADPS can help educators and SENCOs offer timely interventions and support to students at risk of dyslexia, leading to improved academic outcomes and overall well-being. Finally, the gamified design of the ADPS encourages student motivation and self-directed learning, fostering a positive learning environment and reducing the stigma associated with dyslexia assessment.

F. Resolve Resource Scarcity

In Hong Kong, SENCOs have limited resources, and some of them come from non-Chinese backgrounds, which may result in a lack of awareness of reading and writing difficulties in the Chinese language. With this tool, students only need to play four sets of games to help SENCOs understand their performance in different areas of the Chinese language, which speeds up the identification of students at risk of dyslexia.

G. Spotting Chinese Gifted Students With Dyslexia

In addition to identifying students with reading and writing difficulties, this tool can help teachers spot Chinese gifted students with dyslexia. Often, these students have poor grades due to writing difficulties, but they have excellent skills in communication or organization in the Chinese language. With this tool, teachers can provide more appropriate support and assistance, such as providing writing aids or adjusting learning and assessment methods, so that these students can better realize their potential and talents. At the same time, this can help break the stereotype of Chinese gifted students and allow more people to understand their diversity and uniqueness.

H. Limitations

We list several limitations of this study. First, students should sit apart so as not to influence the findings of the prescreening. However, the primary school classrooms in Hong Kong are small, limiting the number of participants. Second, some school infrastructures are insufficient. Due to cloud data uploading and analysis, the prescreening time may increase slightly if the poor Wi-Fi signal quality influences the students' mood. Third, in the reading fluency game, students were suggested to read aloud. However, the students were wearing masks due to the COVID-19 pandemic, making it difficult to collect their voices. Finally, some students told us that the questions from Set 3 were too difficult. The mean score of the experimental group in Set 3 was relatively low, and the scores were most diverse among four sets of prescreening tests (M = 61.443 and SD = 16.123). Furthermore, the experimental group spent the longest time in Set 3 (i.e., M = 839.27 and SD = 220.075). However, the increase in difficulty levels of the questions is inevitable. We may rearrange the order of the four sets of games and mix questions of different difficulty levels.

VI. CONCLUSION

This study introduced the ADPS, an efficient prescreening tool for identifying dyslexia in Traditional Chinese. Using six linguistic characteristics for prescreening yielded a more comprehensive assessment than relying on a single characteristic or nonlinguistic-based screening. Furthermore, incorporating user-friendly design elements, such as TTS and touchscreen-enabled user interfaces, engaged students to complete the prescreening independently without encountering technological literacy issues.

The user study showed that students with dyslexia tend to exhibit weaknesses in VK, MA, and OA. Future research should prioritize improving students' linguistic ability and start with VK to establish a solid foundation for students. Moreover, students with dyslexia often struggle with WM. Therefore, early intervention by SENCOs should focus on enhancing WM through visual training [106].

In the future, we will extend the ADPS to a larger audience. In particular, all the games in the ADPS are transferable to other languages, such as Simplified Chinese, Japanese, and Korean, with minor adjustments in the UI design. Furthermore, given that the strengths and weaknesses of students can be identified through the prescreening tool, this work could be extended to a training tool that provides personalized training for students.

ACKNOWLEDGMENT

The authors would like to thank the anonymous reviewers for their constructive comments.

REFERENCES

- M. J. Snowling and C. Hulme, "Annual research review: Reading disorders revisited—The critical importance of oral language," *J. Child Psychol. Psychiatry*, vol. 62, no. 5, pp. 635–653, 2021.
- [2] K. Y. Fung, K. F. Sin, Z. A. Wen, L.-H. Lee, S. Song, and H. Qu, "Designing a game for pre-screening students with specific learning disabilities in Chinese," in *Proc. 24th Int. ACM SIGACCESS Conf. Comput. Accessibility*, 2022, pp. 1–5.
- [3] R. Burden, "Is dyslexia necessarily associated with negative feelings of self-worth? A review and implications for future research," *Dyslexia*, vol. 14, no. 3, pp. 188–196, 2008.
- [4] E. M. Livingston, L. S. Siegel, and U. Ribary, "Developmental dyslexia: Emotional impact and consequences," *Aust. J. Learn. Difficulties*, vol. 23, no. 2, pp. 107–135, 2018.
- [5] V. W. Berninger et al., "Tier 1 and Tier 2 early intervention for handwriting and composing," J. Sch. Psychol., vol. 44, no. 1, pp. 3–30, 2006.
- [6] A. Brennan, T. McDonagh, M. Dempsey, and J. McAvoy, "Cosmic sounds: A game to support phonological awareness skills for children with dyslexia," *IEEE Trans. Learn. Technol.*, vol. 15, no. 3, pp. 301–310, Jun. 2022.
- [7] K. Y. Fung, Z. A. Wen, H. Li, X. Wang, S. Song, and H. Qu, "Designing a data visualization dashboard for pre-screening Hong Kong students with specific learning disabilities," in *Proc. 24th Int. ACM SIGACCESS Conf. Comput. Accessibility*, 2022, pp. 1–5.
- [8] D. W. Chan, C. S.-H. Ho, S.-M. Tsang, S.-H. Lee, and K. K. Chung, "Reading-related behavioral characteristics of Chinese children with dyslexia: The use of the teachers' behavior checklist in Hong Kong," *Ann. Dyslexia*, vol. 53, pp. 300–323, 2003.
- [9] C. Suk-Han Ho, M.-T. Leung, and H. Cheung, "Early difficulties of Chinese preschoolers at familial risk for dyslexia: Deficits in oral language, phonological processing skills, and print-related skills," *Dyslexia*, vol. 17, no. 2, pp. 143–164, 2011.
- [10] B. H.-H. Ching, C. S.-H. Ho, D. W. Chan, K. K. Chung, and L.-Y. Lo, "Behavioral characteristics of Chinese adolescents with dyslexia: The use of teachers' behavior checklist in Hong Kong," *Appl. Psycholinguistics*, vol. 35, no. 6, pp. 1235–1257, 2014.
- [11] G. Papanastasiou, A. Drigas, C. Skianis, and M. D. Lytras, "Serious games in k-12 education: Benefits and impacts on students with attention, memory and developmental disabilities," *Program*, vol. 51, no. 4, pp. 424–440, 2017.

- [12] D. Lamprinou and F. Paraskeva, "Gamification design framework based on SDT for student motivation," in *Proc. Int. Conf. Interact. Mobile Commun. Technol. Learn.*, 2015, pp. 406–410.
- [13] K. Y. Fung and K. C. Fung, "HCI technology with mastery learning approach for children learning Chinese characters writing in Hong Kong," in Proc. IEEE 20th Int. Conf. Adv. Learn. Technol., 2020, pp. 226–227.
- [14] C. Mejia, B. Florian, R. Vatrapu, S. Bull, S. Gomez, and R. Fabregat, "A novel web-based approach for visualization and inspection of reading difficulties on university students," *IEEE Trans. Learn. Technol.*, vol. 10, no. 1, pp. 53–67, Jan.–Mar. 2017.
- [15] L. Pang and C. C. Jen, "Inclusive dyslexia-friendly collaborative online learning environment: Malaysia case study," *Educ. Inf. Technol.*, vol. 23, pp. 1023–1042, 2018.
- [16] E. Ferrer, B. A. Shaywitz, J. M. Holahan, K. Marchione, and S. E. Shaywitz, "Uncoupling of reading and IQ over time: Empirical evidence for a definition of dyslexia," *Psychol. Sci.*, vol. 21, no. 1, pp. 93–101, 2010.
- [17] C. S.-H. Ho, D. W.-O. Chan, S.-H. Lee, S.-M. Tsang, and V. H. Luan, "Cognitive profiling and preliminary subtyping in Chinese developmental dyslexia," *Cognition*, vol. 91, no. 1, pp. 43–75, 2004.
- [18] K. K. Chung and C. S.-H. Ho, "Dyslexia in Chinese language: An overview of research and practice," *Aust. J. Learn. Difficulties*, vol. 15, no. 2, pp. 213–224, 2010.
- [19] A. Cheng-Lai, C. W. Li-Tsang, A. H. Chan, and A. G. Lo, "Writing to dictation and handwriting performance among Chinese children with dyslexia: Relationships with orthographic knowledge and perceptualmotor skills," *Res. Develop. Disabilities*, vol. 34, no. 10, pp. 3372–3383, 2013.
- [20] C. McBride-Chang, F. Lam, C. Lam, S. Doo, S. W. Wong, and Y. Y. Chow, "Word recognition and cognitive profiles of Chinese pre-school children at risk for dyslexia through language delay or familial history of dyslexia," *J. Child Psychol. Psychiatry*, vol. 49, no. 2, pp. 211–218, 2008.
- [21] S. W. Wong, C. McBride-Chang, C. Lam, B. Chan, F. W. Lam, and S. Doo, "The joint effects of risk status, gender, early literacy and cognitive skills on the presence of dyslexia among a group of high-risk Chinese children," *Dyslexia*, vol. 18, no. 1, pp. 40–57, 2012.
- [22] A. Huang, K. Wu, A. Li, X. Zhang, Y. Lin, and Y. Huang, "The reliability and validity of an assessment tool for developmental dyslexia in Chinese children," *Int. J. Environ. Res. Public Health*, vol. 17, no. 10, 2020, Art. no. 3660.
- [23] Y. Lin et al., "The prevalence of dyslexia in primary school children and their Chinese literacy assessment in Shantou, China," *Int. J. Environ. Res. Public Health*, vol. 17, no. 19, 2020, Art. no. 7140.
- [24] C. McBride-Chang, K. K. Chung, and X. Tong, "Copying skills in relation to word reading and writing in Chinese children with and without dyslexia," *J. Exp. Child Psychol.*, vol. 110, no. 3, pp. 422–433, 2011.
- [25] T. Liu and J. Hsiao, "The perception of simplified and traditional Chinese characters in the eye of simplified and traditional Chinese readers," in *Proc. Annu. Meeting Cogn. Sci. Soc.*, 2012, vol. 34, no. 34, pp. 689–694.
- [26] Y. S. Wei, "A study of modern simplified Chinese characters," Universiti Tunku Abdul Rahman, 2013. [Online]. Available: http://eprints.utar.edu. my/825/
- [27] S. E. Shaywitz and B. A. Shaywitz, "Dyslexia (specific reading disability)," *Pediatrics Rev.*, vol. 24, no. 5, pp. 147–153, 2003.
- [28] M. Rauschenberger, R. Baeza-Yates, and L. Rello, "Screening risk of dyslexia through a web-game using language-independent content and machine learning," in *Proc. 17th Int. Web All Conf.*, 2020, pp. 1–12.
- [29] H. M. Ekhsan, S. Z. Ahmad, S. A. Halim, J. N. Hamid, and N. H. Mansor, "The implementation of interactive multimedia in early screening of dyslexia," in *Proc. Int. Conf. Innov. Manage. Technol. Res.*, 2012, pp. 566–569.
- [30] S. S. Lam, R. K. Au, H. W. Leung, and C. W. Li-Tsang, "Chinese handwriting performance of primary school children with dyslexia," *Res. Develop. Disabilities*, vol. 32, no. 5, pp. 1745–1756, 2011.
- [31] H. Shu, C. McBride-Chang, S. Wu, and H. Liu, "Understanding Chinese developmental dyslexia: Morphological awareness as a core cognitive construct," *J. Educ. Psychol.*, vol. 98, no. 1, 2006, Art. no. 122.
- [32] L. Rello, R. Baeza-Yates, A. Ali, J. P. Bigham, and M. Serra, "Predicting risk of dyslexia with an online gamified test," *Plos One*, vol. 15, no. 12, 2020, Art. no. e0241687.
- [33] M. Rauschenberger, C. Lins, N. Rousselle, A. Hein, and S. Fudickar, "Designing a new puzzle app to target dyslexia screening in pre-readers,"

in Proc. 5th EAI Int. Conf. Smart Objects Technol. Social Good, 2019, pp. 155–159.

- [34] L. Rello et al., "Screening dyslexia for English using HCI measures and machine learning," in *Proc. 2018 Int. Conf. Digit. Health*, 2018, pp. 80–84.
- [35] M. R. U. Saputra and M. Risqi, "LexiPal: Design, implementation and evaluation of gamification on learning application for dyslexia," *Int. J. Comput. Appl.*, vol. 131, no. 7, pp. 37–43, 2015.
- [36] A. De Santo et al., "Promoting computational thinking skills in noncomputer-science students: Gamifying computational notebooks to increase student engagement," *IEEE Trans. Learn. Technol.*, vol. 15, no. 3, pp. 392–405, Jun. 2022.
- [37] A. M. Toda et al., "Analysing gamification elements in educational environments using an existing gamification taxonomy," *Smart Learn. Environ.*, vol. 6, no. 1, 2019, Art. no. 16.
- [38] T. Auvinen, L. Hakulinen, and L. Malmi, "Increasing students' awareness of their behavior in online learning environments with visualizations and achievement badges," *IEEE Trans. Learn. Technol.*, vol. 8, no. 3, pp. 261–273, Jul.–Sep. 2015.
- [39] S. Subhash and E. A. Cudney, "Gamified learning in higher education: A systematic review of the literature," *Comput. Hum. Behav.*, vol. 87, pp. 192–206, 2018.
- [40] L. Rodrigues, A. M. Toda, W. Oliveira, P. T. Palomino, J. Vassileva, and S. Isotani, "Automating gamification personalization to the user and beyond," *IEEE Trans. Learn. Technol.*, vol. 15, no. 2, pp. 199–212, Apr. 2022.
- [41] C. A. Ongoro and Y.-Y. Fangjiang, "Digital game-based technology for English language learning in preschools and primary schools: A systematic analysis," *IEEE Trans. Learn. Technol.*, vol. 17, pp. 202–228, 2023.
- [42] Y. Wang, P. Rajan, C. S. Sankar, and P. Raju, "Let them play: The impact of mechanics and dynamics of a serious game on student perceptions of learning engagement," *IEEE Trans. Learn. Technol.*, vol. 10, no. 4, pp. 514–525, Oct.–Dec. 2017.
- [43] E. Pesare, T. Roselli, N. Corriero, and V. Rossano, "Game-based learning and gamification to promote engagement and motivation in medical learning contexts," *Smart Learn. Environ.*, vol. 3, pp. 1–21, 2016.
- [44] E. Lavoué, B. Monterrat, M. Desmarais, and S. George, "Adaptive gamification for learning environments," *IEEE Trans. Learn. Technol.*, vol. 12, no. 1, pp. 16–28, Jan.–Mar. 2019.
- [45] T. Kucher, "Principles and best practices of designing digital gamebased learning environments," *Int. J. Technol. Educ. Sci.*, vol. 5, no. 2, pp. 213–223, 2021.
- [46] D. Gooch, A. Vasalou, L. Benton, and R. Khaled, "Using gamification to motivate students with dyslexia," in *Proc. CHI Conf. Hum. Factors Comput. Syst.*, 2016, pp. 969–980.
- [47] W. H.-Y. Huang and D. Soman, "Gamification of education," *Rep. Ser.*: *Behav. Econ. Action*, vol. 29, no. 4, 2013, Art. no. 37.
- [48] I. Glover, "Play as you learn: Gamification as a technique for motivating learners," in *EdMedia+ Innovate Learning*. Waynesville, NC, USA: Assoc. Adv. Comput. Educ., 2013, pp. 1999–2008.
- [49] M. Pino and L. Mortari, "The inclusion of students with dyslexia in higher education: A systematic review using narrative synthesis," *Dyslexia*, vol. 20, no. 4, pp. 346–369, 2014.
- [50] G. Elbeheri, Motivating Students with Dyslexia: 100 Ideas for Empowering Teachers in the Classroom. Evanston, IL, USA: Routledge, 2020.
- [51] T. Panitz and P. Panitz, "Encouraging the use of collaborative learning in higher education," in *University Teaching: International Perspectives*. Evanston, IL, USA: Routledge, 1998, pp. 161–201.
- [52] K.-Y. Fung, L.-H. Lee, P. Hui, and S. Song, "Can under- privileged children learn effectively at home? A six-month study of game-based traditional Chinese learning during the pandemic lockdown," *IEEE Trans. Learn. Technol.*, vol. 17, pp. 294–309, 2024.
- [53] Y. Qian, Y.-W. Song, J. Zhao, and H.-Y. Bi, "The developmental trend of orthographic awareness in Chinese preschoolers," *Reading Writing*, vol. 28, no. 4, pp. 571–586, 2015.
- [54] G. K. Georgiou, D. Martinez, A. P. A. Vieira, and K. Guo, "Is orthographic knowledge a strength or a weakness in individuals with dyslexia? Evidence from a meta-analysis," *Ann. Dyslexia*, vol. 71, no. 1, pp. 5–27, 2021.
- [55] Y. K. Wong, "The relationship between orthographic awareness and Chinese reading in Hong Kong's young CSL learners," *ECNU Rev. Educ.*, vol. 3, no. 4, pp. 678–693, 2020.

1468

- [56] C. McBride-Chang, H. Cheung, B.-Y. Chow, C.-L. Chow, and L. Choi, "Metalinguistic skills and vocabulary knowledge in Chinese (11) and English (12)," *Reading Writing*, vol. 19, no. 7, pp. 695–716, 2006.
- [57] C. Suk-Han Ho, T.-T. Ng, and W.-K. Ng, "A 'radical' approach to reading development in Chinese: The role of semantic radicals and phonetic radicals," J. Lit. Res., vol. 25, no. 3, pp. 849–878, 2003.
- [58] J. F. Carlisle, "Effects of instruction in morphological awareness on literacy achievement: An integrative review," *Reading Res. Quart.*, vol. 45, no. 4, pp. 464–487, 2010.
- [59] C. Elbro and E. Arnbak, "The role of morpheme recognition and morphological awareness in dyslexia," *Ann. Dyslexia*, vol. 46, pp. 209–240, 1996.
- [60] H. Oz, "Morphological awareness and some implications for English language teaching," *Procedia—Social Behav. Sci.*, vol. 136, pp. 98–103, 2014.
- [61] S. A. Storch and G. J. Whitehurst, "Oral language and coderelated precursors to reading: Evidence from a longitudinal structural model," *Develop. Psychol.*, vol. 38, no. 6, 2002, Art. no. 934.
- [62] M. Snowling, "Dyslexia as a phonological deficit: Evidence and implications," *Child Psychol. Psychiatry Rev.*, vol. 3, no. 1, pp. 4–11, 1998.
- [63] J. Pan et al., "On the relationship between phonological awareness, morphological awareness and Chinese literacy skills: Evidence from an 8-year longitudinal study," *Develop. Sci.*, vol. 19, no. 6, pp. 982–991, 2016.
- [64] Y. Ye, C. K.-Y. Tong, C. McBride, L. Yin, and C. Y. Tse, "Concurrent and longitudinal predictors of beginning writing in Chinese: The unique role of copying skills," *Reading Writing*, vol. 34, no. 9, pp. 2417–2437, 2021.
- [65] C. A. Knoop-van Campen, E. Segers, and L. Verhoeven, "How phonological awareness mediates the relation between working memory and word reading efficiency in children with dyslexia," *Dyslexia*, vol. 24, no. 2, pp. 156–169, 2018.
- [66] B. W.-Y. Chow, C. McBride-Chang, and S. Burgess, "Phonological processing skills and early reading abilities in Hong Kong Chinese kindergarteners learning to read English as a second language," *J. Educ. Psychol.*, vol. 97, no. 1, pp. 81–87, 2005.
- [67] X. Tong and C. McBride-Chang, "Developmental models of learning to read Chinese words," *Develop. Psychol.*, vol. 46, no. 6, 2010, Art. no. 1662.
- [68] A. M.-Y. Wong, J. C. Kidd, C. S.-H. Ho, and T. K.-F. Au, "Characterizing the overlap between SLI and dyslexia in Chinese: The role of phonology and beyond," *Sci. Stud. Reading*, vol. 14, no. 1, pp. 30–57, 2010.
- [69] I. Lieder, V. Adam, O. Frenkel, S. Jaffe-Dax, M. Sahani, and M. Ahissar, "Perceptual bias reveals slow-updating in autism and fast-forgetting in dyslexia," *Nature Neurosci.*, vol. 22, no. 2, pp. 256–264, 2019.
- [70] E. Sumner, V. Connelly, and A. L. Barnett, "Children with dyslexia are slow writers because they pause more often and not because they are slow at handwriting execution," *Reading Writing*, vol. 26, no. 6, pp. 991–1008, 2013.
- [71] M. Martlewm, "Handwriting and spelling: Dyslexic children's abilities compared with children of the same chronological age and younger children of the same spelling level," *Brit. J. Educ. Psychol.*, vol. 62, no. 3, pp. 375–390, 1992.
- [72] K. Y. Fung and K. Y. Tang, "Multi-modal interactive dyslexia, dysgraphia and dyspraxia classification system," U.S. Patent 63/255 962, Oct. 2021.
- [73] K. Y. Fung, K.-C. Fung, and S.-Y. Wan, "Augmented reality and 3D model for children Chinese character recognition-Hong Kong primary school education," in *Proc. 27th Int. Conf. Comput. Educ.*, 2019, vol. 1, pp. 673–678.
- [74] K. Y. Fung, C. Kwong, C. Aidan, and Y. Y. Ching, "A digital tool to provide pre-screening to dyslexia in Hong Kong," in *Proc. IEEE Int. Conf. Eng., Technol. Educ.*, 2021, pp. 755–761.
- [75] N. Gregg, C. Coleman, M. Davis, and J. C. Chalk, "Timed essay writing: Implications for high-stakes tests," *J. Learn. Disabilities*, vol. 40, no. 4, pp. 306–318, 2007.
- [76] J. Hatcher, M. J. Snowling, and Y. M. Griffiths, "Cognitive assessment of dyslexic students in higher education," *Brit. J. Educ. Psychol.*, vol. 72, no. 1, pp. 119–133, 2002.
- [77] L. B. Feldman and W. W. Siok, "The role of component function in visual recognition of Chinese characters," J. Exp. Psychol.: Learn., Memory, Cogn., vol. 23, no. 3, pp. 776–781, 1997.
- [78] L. H. Tan, J. A. Spinks, G. F. Eden, C. A. Perfetti, and W. T. Siok, "Reading depends on writing, in Chinese," *Proc. Nat. Acad. Sci.*, vol. 102, no. 24, pp. 8781–8785, 2005.

- [79] S. C. Kalindi, C. McBride, X. Tong, N. L. Y. Wong, K. H. K. Chung, and C.-Y. Lee, "Beyond phonological and morphological processing: Pure copying as a marker of dyslexia in Chinese but not poor reading of English," *Ann. Dyslexia*, vol. 65, no. 2, pp. 53–68, 2015.
- [80] A. D. Baddeley, "Is working memory still working?," *Eur. Psychol.*, vol. 7, no. 2, pp. 85–97, 2002.
- [81] J. H. Smith-Spark and J. E. Fisk, "Working memory functioning in developmental dyslexia," *Memory*, vol. 15, no. 1, pp. 34–56, 2007.
- [82] H. L. Swanson, J. Mink, and K. M. Bocian, "Cognitive processing deficits in poor readers with symptoms of reading disabilities and ADHD: More alike than different?," *J. Educ. Psychol.*, vol. 91, no. 2, pp. 321–333, 1999.
- [83] C. Maehler and K. Schuchardt, "Working memory in children with specific learning disorders and/or attention deficits," *Learn. Individual Differ.*, vol. 49, pp. 341–347, 2016.
- [84] W. Tops, C. Callens, E. Van Cauwenberghe, J. Adriaens, and M. Brysbaert, "Beyond spelling: The writing skills of students with dyslexia in higher education," *Reading Writing*, vol. 26, no. 5, pp. 705–720, 2013.
- [85] V. W. Berninger, W. Raskind, T. Richards, R. Abbott, and P. Stock, "A multidisciplinary approach to understanding developmental dyslexia within working-memory architecture: Genotypes, phenotypes, brain, and instruction," *Develop. Neuropsychol.*, vol. 33, no. 6, pp. 707–744, 2008.
- [86] L. T. Rose and P. Rouhani, "Influence of verbal working memory depends on vocabulary: Oral reading fluency in adolescents with dyslexia," *Mind, Brain, Educ.*, vol. 6, no. 1, pp. 1–9, 2012.
- [87] G. P. Ouellette, "What's meaning got to do with it: The role of vocabulary in word reading and reading comprehension," *J. Educ. Psychol.*, vol. 98, no. 3, 2006, Art. no. 554.
- [88] N. Schmitt, "Size and depth of vocabulary knowledge: What the research shows," *Lang. Learn.*, vol. 64, no. 4, pp. 913–951, 2014.
- [89] Y. Zhou et al., "Development of reading-related skills in Chinese and English among Hong Kong Chinese children with and without dyslexia," *J. Exp. Child Psychol.*, vol. 122, pp. 75–91, 2014.
- [90] M. D. C. Tongco, "Purposive sampling as a tool for informant selection," *Res. Appl.*, vol. 5, pp. 147–158, 2007.
- [91] K.-Y. Fung, S. T. Perrault, L.-H. Lee, K.-C. Fung, and S. Song, "Can students with dyslexia learn independently? A seven-week study of Chinese character learning in an informal learning environment," *IEEE Trans. Learn. Technol.*, vol. 16, no. 5, pp. 619–632, Oct. 2023.
- [92] Y. Yang et al., "The brain basis of handwriting deficits in Chinese children with developmental dyslexia," *Develop. Sci.*, vol. 25, no. 2, 2022, Art. no. e13161.
- [93] M. S. A. B. Rosli, I. S. Isa, S. A. Ramlan, S. N. Sulaiman, and M. I. F. Maruzuki, "Development of CNN transfer learning for dyslexia handwriting recognition," in *Proc. IEEE 11th Int. Conf. Control Syst., Comput. Eng.*, 2021, pp. 194–199.
- [94] N. S. L. Seman, I. S. Isa, S. A. Ramlan, W. Li-Chih, and M. I. F. Maruzuki, "Classification of handwriting impairment using CNN for potential dyslexia symptom," in *Proc. IEEE 11th Int. Conf. Control Syst.*, *Comput. Eng.*, 2021, pp. 27–28.
- [95] M. Volman, B. M. van Schendel, and M. J. Jongmans, "Handwriting difficulties in primary school children: A search for underlying mechanisms," *Amer. J. Occupat. Ther.*, vol. 60, no. 4, pp. 451–460, 2006.
- [96] S. Kohnen, L. Nickels, A. Castles, N. Friedmann, and G. McArthur, "When 'slime' becomes 'smile': Developmental letter position dyslexia in English," *Neuropsychologia*, vol. 50, no. 14, pp. 3681–3692, 2012.
- [97] C. Juel, "Learning to read and write: A longitudinal study of 54 children from first through fourth grades," *J. Educ. Psychol.*, vol. 80, no. 4, pp. 437–447, 1988.
- [98] D. Menghini, A. Finzi, G. A. Carlesimo, and S. Vicari, "Working memory impairment in children with developmental dyslexia: Is it just a phonological deficity?," *Develop. Neuropsychol.*, vol. 36, no. 2, pp. 199–213, 2011.
- [99] K. Landerl, U. Frith, and H. Wimmer, "Intrusion of orthographic knowledge on phoneme awareness: Strong in normal readers, weak in dyslexic readers," *Appl. Psycholinguistics*, vol. 17, no. 1, pp. 1–14, 1996.
- [100] L.-J. Kuo and R. C. Anderson, "Morphological awareness and learning to read: A cross-language perspective," *Educ. Psychol.*, vol. 41, no. 3, pp. 161–180, 2006.
- [101] R. Ranchal et al., "Using speech recognition for real-time captioning and lecture transcription in the classroom," *IEEE Trans. Learn. Technol.*, vol. 6, no. 4, pp. 299–311, Oct.–Dec. 2013.
- [102] M. Weerasinghe et al., "VocabulARy: Learning vocabulary in AR supported by keyword visualisations," *IEEE Trans. Vis. Comput. Graph.*, vol. 28, no. 11, pp. 3748–3758, Nov. 2022.

- [103] D. Hawes and A. Arya, "Technology solutions to reduce anxiety and increase cognitive availability in students," *IEEE Trans. Learn. Technol.*, vol. 16, no. 2, pp. 278–291, Apr. 2023.
- [104] L. A. Annetta, J. Minogue, S. Y. Holmes, and M.-T. Cheng, "Investigating the impact of video games on high school students' engagement and learning about genetics," *Comput. Educ.*, vol. 53, no. 1, pp. 74–85, 2009.
- [105] R. Garris, R. Ahlers, and J. E. Driskell, "Games, motivation, and learning: A research and practice model," *Simul. Gaming*, vol. 33, no. 4, pp. 441–467, 2002.
- [106] C. Hulme and M. J. Snowling, "The interface between spoken and written language: Developmental disorders," *Philos. Trans. Roy. Soc. B: Biol. Sci.*, vol. 369, no. 1634, 2014, Art. no. 20120395.



Ka-Yan Fung received the M.Phil. degree in technology, leadership, and entrepreneurship from the Hong Kong University of Science and Technology, Hong Kong, in 2021, and the M.Phil. degree in contemporary east Asian studies from the University of Duisburg-Essen, Duisburg, Germany, in 2021. She is currently working toward the Ph.D. degree in human–computer interaction with the Individualized Interdisciplinary Program, Hong Kong University of Science and Technology.

Her research interests include special learning disabilities, human–computer interaction, human–robot interaction, data visualization, and inclusive education.



Kit-Yi Tang received the Ph.D. degree in psychological education from the Chinese University of Hong Kong, Hong Kong, in 2009.

Before her retirement, she dedicated herself to teacher training and inclusive education in different universities, e.g., Hong Kong Baptist University, Hong Kong, Hong Kong Education University, Hong Kong, and the Chinese University of Hong Kong, Hong Kong. In addition, to help special education needs (SEN) students of different age groups learn better, she has improved different teaching methods,

and also participated in the development of board games (for young adults with autism) and electronic rapid testing tools (for preschool children and junior students may have the problem in reading and writing).



Tze Leung Rick Lui received the Ed.D. degree in dyslexic identity of adolescents from the Education University of Hong Kong, Hong Kong, in 2019.

He is currently a Project Manager with the Centre for Special Educational Needs and Inclusive Education, Education University of Hong Kong, responsible for the SMILE Extension project and teachers' training programs. His research interests include catering for students of gifted education and twice exceptional (gifted with learning difficulties).



Kuen-Fung Sin received the Doctor of Education degree in special education from Leicester University, Leicester, U.K., in 2001.

He is currently the Executive Director of the Institute of Special Needs and Inclusive Education, Tai Po, Hong Kong, and a Professor with the Department of Special Education and Counselling, Education University of Hong Kong (EdUHK), Hong Kong. He takes part in many community activities, in relation to special needs and inclusion. His research focuses on supporting students with special educational needs

and professional development in inclusion.

Dr. Sin was the recipient of several awards, including the President's Awards for Outstanding Performance in Knowledge Transfer (Individual), Teaching (Individual), Knowledge Transfer (Team), and Administrative Services (Individual) from EdUHK in 2017–2020 and 2022. In 2021, he received the Medal of Honor from the government of Hong Kong in recognition of his achievement in developing special education, particularly his contribution to enhancing special education training for teachers.



Lik-Hang Lee received the bachelor's degree in logistics engineering and supply chain management, in 2011, the M.Phil. degree in industrial and manufacturing systems engineering from the University of Hong Kong, Hong Kong, in 2014, and the Ph.D. degree in computer science and engineering from the Hong Kong University of Science and Technology, Hong Kong, in 2019.

He is currently an Assistant Professor with Hong Kong Polytechnic University, Hong Kong. From 2019 to 2020, he was Postdoctoral Researcher at the Uni-

versity of Oulu, Oulu, Finland. From 2021 to 2023, he was an Assistant Professor (tenure-track) at the Korea Advanced Institute of Science and Technology, Daejeon, South Korea. He has built and designed various human-centric computing systems, specializing in augmented and virtual realities. He has served the research community at well-recognized conferences and journals in recent years, such as the AAAI Conference on Artificial Intelligence, the International Joint Conference on Artificial Intelligence, the IEEE International Conference on Pervasive Computing and Communications, the ACM Conference on Human Factors in Computing Systems, the International World Wide Web Conference, ACM Multimedia, *Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies, ACM Computing Surveys*, and the IEEE Conference on Virtual Reality and 3D User Interfaces.



Huamin Qu (Member, IEEE) received the Ph.D. degree in computer science from Stony Brook University, Stony Brook, NY, USA, in 2004.

He is currently a Dean and Chair Professor with the Hong Kong University of Science and Technology, Hong Kong. He has coauthored more than 200 publications. His research interests include data visualization, human–computer interaction, and E-learning.

Dr. Qu is a recipient of the IEEE VGTC Visualization Technical Achievement Award. He was a Founding Member of the IEEE VIS Steering Committee.

He was an Associate Editor for IEEE TRANSACTIONS ON VISUALIZATION AND COMPUTER GRAPHICS. He was inducted into the IEEE Visualization Academy in 2020.



Shenghui Song (Senior Member, IEEE) received the M. Eng. degree from Tianjin University, Tianjin, China, in 2002, and the Ph.D. degree from the City University of Hong Kong, Hong Kong, in 2006.

He is an Assistant Professor with the Division of Integrative Systems and Design and the Department of Electronic and Computer Engineering, Hong Kong University of Science and Technology (HKUST), Hong Kong. His research interests include wireless communications and machine learning with current focus on distributed intelligence, machine learning

for communications, integrated sensing and communication, and information theory as well as engineering education.

Dr. Song was an Exemplary Reviewer for IEEE COMMUNICATIONS LETTERS. He was the Tutorial Program Co-Chair of the 2022 IEEE International Mediterranean Conference on Communications and Networking. He was an Associate Editor for IEEE TRANSACTIONS ON EDUCATION from 2013 to 2022. He has received several teaching awards from HKUST, including the Michael G. Gale Medal for Distinguished Teaching in 2018, the Best Ten Lecturers in 2013, 2015, and 2017, the School of Engineering Distinguished Teaching Award in 2012, the Teachers I Like Award in 2013, 2015, 2016, and 2017, and the M.Sc. (Telecom) Teaching Excellent Appreciation Award in 2021. He was one of the honorees of the Third Faculty Recognition at HKUST in 2021.